Author's response to reviews

Title: Simulated learning in musculoskeletal assessment and rehabilitation education: Exploring its benefits on skills and confidence development, as well as motivation for future learning

Authors:

Mark D Hecimovich (m.hecimovich@murdoch.edu.au)
Simone E Volet (s.volet@murdoch.edu.au)

Version: 2 Date: 23 July 2014

Author's response to reviews:

Dear Editor,

We have formatted the manuscript per BMC requirements and suggestions. The Abstract has been edited, line numbering added, and title page with author information attached to the main manuscript. I am hopeful we have met these requirements and the manuscript can be placed into peer review. Thank you very much.

Regards,

Mark Hecimovich